Author’s response to reviews

Title: Comparison of a loading dose of dexmedetomidine combined with propofol or sevoflurane for hemodynamic changes during anesthesia maintenance: a prospective, randomized, double blind, controlled clinical trial

Authors:

Yuan Han (hanyuan_trial@163.com)
Liu Han (hanliu201009@163.com;334903226@qq.com)
Meng Dong (dongmengmcxz@126.com)
Qing Sun (sunqingchun2010@163.com)
Ke Ding (dke2010@163.com)
Zhen Zhang (zzfeng_88@163.com)
Jun Cao (junlicao0310@163.com)
Yue Zhang (zhangyy201009@163.com)

Version: 4 Date: 29 Dec 2017

Author’s response to reviews:

Dear Editor,

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript, and we also appreciate you very much for your constructive suggestions on our manuscript entitled our manuscript entitled “A comparison of a loading dose of Dexmedetomidine combined with Propofol or Sevoflurane for hemodynamic changes during anesthesia maintenance: a prospective, randomized, double-blind, controlled clinical trial" (BANE-D-17-00263).

We have studied your comments very carefully and have made revision. According to your kind suggestion, we added email address of all authors, the date of trial registration and removed all
the tracked changes in our article. And also, we described the detail contribution of each author. Attached please find the revised version.

We would like to express our great appreciation to you for comments on our paper again. It deeply helps us enrich the content of the article. Looking forward to hearing from you.

Thank you and best regards.

Yours sincerely,

Yueying Zhang, M.D.
Corresponding author

E-mail: zhangyy201009@163.com